Author’s response to reviews

Title: Evaluating cutpoints for the MHI-5 and MCS using the GHQ-12: A comparison of five different methods

Authors:

Mark J Kelly (kellymj1@cf.ac.uk)
Frank D Dunstan (dunstanfd@cf.ac.uk)
Keith Lloyd (k.r.lloyd@swansea.ac.uk)
David L Fone (foned@cf.ac.uk)

Version: 3 Date: 15 February 2008

Author's response to reviews:

Title & Abstract in the submission system and the manuscript are now identical

Affiliation 2 now has a country at the end of it.

Authors contributions has been changed to Authors’ contributions

It has been indicated that all authors have read and approved the final manuscript.

Figure legends have been removed from footnotes and are now included in the text proper.